Author's response to reviews

Title: Homeopathy for Allergic Rhinitis: Protocol for Systematic Review

Authors:

Kushal Banerjee (kushal@drkbanerjee.com)
Ceire Costelloe (c.costelloe@qmul.ac.uk)
Robert T Mathie (r.mathie@britishhomeopathic.org)
Jeremy Howick (jeremy/howick@phc.ox.ac.uk)

Version: 6 Date: 23 May 2014

Author's response to reviews: see over
From,
Dr. Kushal Banerjee
Dr. Kalyan Banerjee’s Clinic
I 1691 Chittaranjan Park
New Delhi 110019
India
kushal@drkbanerjee.com

22\textsuperscript{nd} May 2014

To,
Dr. Mark Rogers,
Editor,
BMC Systematic Reviews

\textbf{Subject: Third Resubmission after review MS: 6829843531197782}

Dear Dr. Rogers,

Thank you for inviting us to resubmit the systematic review protocol, ‘Homeopathy for Allergic Rhinitis’, for a third time, after addressing the suggested addition to the title.

As requested, I am submitting the revised manuscript with changes in track.

Please do not hesitate to contact us for any further clarifications.

Yours sincerely,

Kushal Banerjee*, Ceire Costelloe, Robert Mathie and Jeremy Howick

*Corresponding Author
**Detailed reply to reviewer comments**

<table>
<thead>
<tr>
<th>S. No.</th>
<th>Editor’s Comments</th>
<th>Author Response</th>
</tr>
</thead>
<tbody>
<tr>
<td>1.</td>
<td>I would like the authors to add three words to their title - systematic review protocol.</td>
<td>We understand the need to clarify that this is a systematic review protocol in the title. This change has been made in the manuscript on pages 1 and 2. The title now reads, ‘Homeopathy for Allergic Rhinitis : Protocol for Systematic Review’. This change will also be made in the electronic submission system when the new new manuscript is uploaded.</td>
</tr>
</tbody>
</table>